Author’s response to reviews

Title: Cefuroxime, levofloxacin, esomeprazole, and bismuth as first-line Helicobacter pylori eradication therapy in patients allergic to penicillin

Authors:

Zhiqiang Song (szqinput@163.com)
Wei Fu (weierhsc@qq.com)
Liya Zhou (zhouliya123456@163.com)

Version: 1 Date: 26 Jan 2019

Author’s response to reviews:

Manuscript ID: BMGE-D-18-00769
Manuscript title: Cefuroxime, levofloxacin, esomeprazole, and bismuth as first-line therapy for eradicating Helicobacter pylori in patients allergic to penicillin

Date: 18 Jan 2019
To: Dr. Reeves

Dear editor,

Thanks so much for your feedback regarding our manuscript -BMGE-D-18-00769 titled “Cefuroxime, levofloxacin, esomeprazole, and bismuth as first-line Helicobacter pylori eradication therapy in patients allergic to penicillin” recently submitted for publication to BMC Gastroenterology.

We have followed your requirements and revised our original manuscript accordingly. Notably, textual overlap was minimized throughout the manuscript. In addition, the headings were modified (Introduction to Background; Materials and Methods to Methods), deleted (“Aim” in the Abstract) or added (Conclusions) as suggested. Author names were corrected to follow your editorial style, and a clause was added to indicate that the funding body did not influence the study.

We confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.
Best wishes,

Dr Zhou